Author's response to reviews

Title: A conceptual framework for implementation fidelity

Authors:

Christopher Carroll (c.carroll@shef.ac.uk)
Malcolm Patterson (m.patterson@shef.ac.uk)
Stephen Wood (s.j.wood@shef.ac.uk)
Andrew Booth (a.booth@shef.ac.uk)
Jo Rick (jo.rick@shef.ac.uk)
Shashi Balain (s.balain@shef.ac.uk)

Version: 5 Date: 5 November 2007

Author's response to reviews:

Dear Professor Eccles,

I have uploaded a new clean copy as requested - the only issue I had with the copyediting was that Figure 1 has been incorrectly ascribed - the "box" that has been designated Figure 1 is in fact simply a box that appeared in the text on p.8 and was not designated a figure at all; the actual Figure 1 was uploaded as a separate document, as required by the submission process. Worryingly, it does not appear on this copy of the document at all - the true Figure 1 has been uploaded again, with the recommended revised, "clean" copy - the text will not make sense without this correct designation of the correct figure.

Thank you.

Kind regards

Christopher Carroll